Author's response to reviews

Title: Feasibility and costs of water fluoridation in remote Australian Aboriginal communities

Authors:

Jonathon P Ehsani (jpehsani@hotmail.com)
Ross Bailie (Ross.Bailie@menzies.edu.au)

Version: 4 Date: 28 May 2007

Author's response to reviews: see over
Dear Anita,

Thank you for the acceptance of our paper (MS: 1979928173123119) on the feasibility and costs of water fluoridation in remote Australian Aboriginal communities. We also enclose the revised manuscript with the change marked using ‘Track Changes’.

Sincerely

Jonathon P. Ehsani and Ross Bailie